\documentstyle[harvard]{article}
\begin{document}
\bibliographystyle{test}
\harvardparenthesis{round}
\citationstyle{dcu}
\citationmode{full}
\renewcommand{\harvardand}{\&}

Some \citeyear{cross80} say there are X.

Some \citeyear{cross80} say there are X.

Some \citeyear{cross79} say there are X.

Some \citeyear{cross79} say there are X.

Some \citeyear{cross81} say there are X.

Some \citeyear{cross81} say there are X.

Some \citeyear{cross76} say there are X.

Some \citeyear{cross76} say there are X.

Some \citename{cross80} say there are X.

Some \citename{cross80} say there are X.

Some \citename{cross79} say there are X.

Some \citename{cross79} say there are X.

Some \citename{cross81} say there are X.

Some \citename{cross81} say there are X.

Some \citename{cross76} say there are X.

Some \citename{cross76} say there are X.

Some \cite{MG-et:1968} say there are X.

Some \cite{cross80} say there are X.

Some \cite{cross80} say there are X.

Some \cite{cross79} say there are X.

Some \cite{cross79} say there are X.

Some \cite{cross81} say there are X.

Some \cite{cross81} say there are X.

Some \cite{cross76} say there are X.

Some \cite{cross76} say there are X.

Some \cite{cross81} say there are X.

Some \cite{cross81} say there are X.

Some \cite{cross76} say there are X.

Some \cite{cross76x} say there are X.

Some \cite{cross81x} say there are X.

Some \cite{cross81x} say there are X.

Some \cite{cross76} say there are X.

Some \cite{cross76} say there are X.

Some \cite{article-minimal} say there are X.

Some \cite{article-full} say there are X.

Some \cite{article-full-b} say there are X.

Some \cite{article-crossref} say there are X.

Some \cite{inbook-minimal} say there are X.

Some \cite{inbook-full,book-full} say there are X.

Some \cite{inbook-crossref} say there are X.

Some \cite{book-minimal} say there are X.

Some \cite{book-full} say there are X.

Some \cite{book-crossref} say there are X.

Some \cite{whole-set} say there are X.

Some \cite{booklet-minimal} say there are X.

Some \cite{booklet-full} say there are X.

Some \cite{incollection-minimal} say there are X.

Some \cite{incollection-full} say there are X.

Some \cite{incollection-crossref} say there are X.

Some \cite{whole-collection} say there are X.

Some \cite{manual-minimal} say there are X.

Some \cite{manual-full} say there are X.

Some \cite{manual-full-b} say there are X.

Some \cite{mastersthesis-minimal} say there are X.

Some \cite{mastersthesis-full} say there are X.

Some \cite{misc-minimal} say there are X.

Some \cite{misc-full} say there are X.

Some \cite{inproceedings-minimal} say there are X.

Some \cite{inproceedings-full} say there are X.

Some \cite{inproceedings-full-b} say there are X.

Some \cite{inproceedings-crossref} say there are X.

Some \cite{proceedings-minimal} say there are X.

Some \cite{proceedings-full} say there are X.

Some \cite{whole-proceedings} say there are X.

Some \cite{phdthesis-minimal} say there are X.

Some \cite{phdthesis-full} say there are X.

Some \cite{techreport-minimal} say there are X.

Some \cite{techreport-full} say there are X.

Some \cite{unpublished-minimal} say there are X.

Some \cite{unpublished-full} say there are X.

Some \cite{random-note-crossref} say there are X.

Some \citeasnoun{article-minimal} say there are X.

Some \citeasnoun{article-full} say there are X.

Some \citeasnoun{article-full-b} say there are X.

Some \citeasnoun{article-crossref} say there are X.

Some \citeasnoun{inbook-minimal} say there are X.

Some \citeasnoun{inbook-full} say there are X.

Some \citeasnoun{inbook-crossref} say there are X.

Some \citeasnoun{book-minimal} say there are X.

Some \citeasnoun{book-full} say there are X.

Some \citeasnoun{book-crossref} say there are X.

Some \citeasnoun{whole-set} say there are X.

Some \citeasnoun{booklet-minimal} say there are X.

Some \citeasnoun{booklet-full} say there are X.

Some \citeasnoun{incollection-minimal} say there are X.

Some \citeasnoun{incollection-full} say there are X.

Some \citeasnoun{incollection-crossref} say there are X.

Some \citeasnoun{whole-collection} say there are X.

Some \citeasnoun{manual-minimal} say there are X.

Some \citeasnoun{manual-full} say there are X.

Some \citeasnoun{manual-full-b} say there are X.

Some \citeasnoun{mastersthesis-minimal} say there are X.

Some \citeasnoun{mastersthesis-full} say there are X.

Some \citeasnoun{misc-minimal} say there are X.

Some \citeasnoun{misc-full} say there are X.

Some \citeasnoun{inproceedings-minimal} say there are X.

Some \citeasnoun{inproceedings-full} say there are X.

Some \citeasnoun{inproceedings-full-b} say there are X.

Some \citeasnoun{inproceedings-crossref} say there are X.

Some \citeasnoun{proceedings-minimal} say there are X.

Some \citeasnoun{proceedings-full} say there are X.

Some \citeasnoun{whole-proceedings} say there are X.

Some \citeasnoun{phdthesis-minimal} say there are X.

Some \citeasnoun{phdthesis-full} say there are X.

Some \citeasnoun{techreport-minimal} say there are X.

Some \citeasnoun{techreport-full} say there are X.

Some \citeasnoun{unpublished-minimal} say there are X.

Some \citeasnoun{unpublished-full} say there are X.

Some \citeasnoun{random-note-crossref} say there are X.

\bibliography{xampl}
\end{document}
